Author's response to reviews

Title: Factors associated with cognition in late childhood and adolescence: the Young Lives cohort study of children in Ethiopia, India, Peru and Vietnam

Authors:

Benjamin T Crookston (benjamin_crookston@byu.edu)
Renata Forste (renata_forste@byu.edu)
Christine McClellan (cm@byu.net)
Andreas Georgiadis (andreas.georgiadis@qeh.ox.ac.uk)
Tim B Heaton (tim_heaton@byu.edu)

Version: 2 Date: 12 May 2014

Author's response to reviews:

Line and page numbers have now been added. We have also indicated in the methods section that written informed consent was provided by each household. We apologize for this oversight.